Author's response to reviews

Title: Factors associated with poor prognosis among patients admitted with heart failure in a Nigerian tertiary medical centre: a cross-sectional study

Authors:

Kamilu M Karaye (kkaraye@yahoo.co.uk)
Mahmoud U Sani (sanimahmoud@yahoo.com)

Version: 8 Date: 19 July 2008

Author's response to reviews:

The following are the final revisions to the manuscript:

1. Discussion section, Page 13, 3rd paragraph, 2nd line: there were 2 commas [,.] one of which was deleted.

2. References section, #33: the year of publication was 1977 and not 1997.

3. In the manuscript, 2 spaces were mistakenly added. These have now been deleted, and for this reason the number of pages of the whole manuscript has now reduced to 32 instead of 33.